Author’s response to reviews

Title: Productive activities, mental health and quality of life in disability: Exploring the role enhancement and the role strain hypotheses

Authors:

Christine Fekete (christine.fekete@paraplegie.ch)
Johannes Siegrist (johannes.siegrist@med.uni-duesseldorf.de)
Marcel Post (m.post@dehoogstraat.nl)
Martin Brinkhof (martin.brinkhof@paraplegie.ch)

Version: 3 Date: 10 Dec 2018

Author’s response to reviews:

We are pleased that we are now close to publication and would like to thank you again for taking the time to handle our manuscript. We have now deposited the data at dryad.

Yours sincerely,

Christine Fekete on behalf of the co-authors, Johannes Siegrist, Marcel Post & Martin Brinkhof

Editor Comments:

1. We kindly ask that at this stage you make your data available. Please deposit your data and provide the link and identifiers to access the data in this section.

We have now uploaded the dataset on dryad and have added the doi:10.5061/dryad.6265g37 in the final version of the manuscript (p. 17, line 440).

2. Please clarify whether the questionnaire used in your study was developed for this study or has previously been published elsewhere. If the questionnaire has been published elsewhere please provide a reference to it in your manuscript, if the questionnaire was developed for this study please upload an English language version as a supplementary file.

All questionnaires used in this study have been previously published. Relevant literature has been cited for all questionnaires in the measures section of the manuscript (p. 6, lines 156ff.).